Author's response to reviews

Title: Antiviral activity of Bifidobacterium adolescentis SPM 1005-A on human papillomavirus type 16

Authors:

Min Kyeong Cha (cmkung1203@naver.com)
Do Kyung Lee (015790@hanmail.net)
Hyang Mi An (aashhm@hanmail.net)
Si Won Lee (minon.elly@gmail.com)
Seon Hee Shin (shshin@hallym.or.kr)
Jeong-Hyun Kwon (youbc@jwu.ac.kr)
Kyung Jae Kim (kimkj@syu.ac.kr)
Nam Joo Ha (hanj@syu.ac.kr)

Version: 7 Date: 25 June 2012

Author's response to reviews: see over
June 25, 2012

BMC Medicine
Dear Editor in Chief

I would very much like to submit a paper to the “Medicine”. Please review our manuscript entitled “Antiviral activity of Bifidobacterium adolescentis SPM 1005-A on human papillomavirus type 16” with your kind consideration. This manuscript has not been published in part and is not being considered for publication elsewhere.

I look forward to hearing good news from you soon.

Sincerely yours,

Ph.D. Nam Joo Ha, Professor
College of Pharmacy
Sahmyook University
Seoul 139-742, Korea
Tel.: 82-2-3399-1607
Fax: 82-2-3399-1617
E-mail: hanj@syu.ac.kr